Author's response to reviews

Title: The CHOICE study: women's perceptions and reasons for choosing the pill, patch or ring

Authors:

Christian Egarter (christian.egarter@meduniwien.ac.at)
Brigitte Frey Tirri (Brigitte.frey@ksbh.ch)
Johannes Bitzer (jbitzer@uhbs.ch)
Vyacheslav Kaminskyy (kamanat@mail.ru)
Björn J Oddens (bjorn.oddens@merck.com)
Vera Prilepskaya (vprilepskaya@mail.ru)
Arie Yeshaya (yeshaya.a.j@012.net.il)
Maya Marintcheva-Petrova (maya.Marintcheva-Petrova@merck.com)
Steven Weyers (steven.weyers@ugent.be)

Version: 2 Date: 19 October 2012

Author's response to reviews: see over
Dear Dr. Crow:

On behalf of my coauthors, I am pleased to resubmit the manuscript titled “The CHOICE study: women’s perceptions and reasons for choosing the pill, patch or ring” for your consideration for publication in the journal *BMC Women’s Health*.

We have made the following changes to the manuscript in response to your 19-October email:

- On page 7 we included text summarising information presented in the primary report of the study (reference 6, Eur J Contracept Reprod Health Care, 2012, 17:65-78), indicating that all women participants of this study provided informed consent as well as details about how and where ethical approvals were obtained in the 11 countries included in the CHOICE study.
- On page 19 we have provided a header for the Conclusions section.
- We have uploaded the 2 figure files in MS Word format (.doc) in portrait configuration.

I hope you find this manuscript to be worthy of publication in *BMC Women’s Health*.

If you have questions, please let me know and I will do my best to assist you.

Thank you for your consideration.

Sincerely,

Christian

Christian Egarter, M.D.
Medical University of Vienna
Department of Gynecological Endocrinology and Reproduction
Währinger Gürtel 18-20
1090 Vienna, Austria
Tel: +43-1-40400-2822
Fax: +43-1-40400-2861
E-mail: christian.egarter@meduniwien.ac.at